Author's response to reviews

Title: CONNECT for Quality: Protocol of a Randomized Controlled Trial to Improve Fall Prevention in Nursing Homes

Authors:

Ruth A Anderson (ruth.anderson@duke.edu)
Kirsten Corazzini (coraz001@mc.duke.edu)
Kristie Porter (Kirstie.porter@duke.edu)
Katherine Daily (katherine.daily@duke.edu)
Reuben R McDaniel Jr. (Reuben.McDaniel@mccombs.utexas.edu)
Cathleen Colon-Emeric (Colon001@mc.duke.edu)

Version: 5 Date: 8 February 2012

Author's response to reviews: see over
Dr. Martin Eccles and Dr. Brian Mittman  
Implementation Science  
BioMed Central Ltd  
Floor 6, 236 Gray's Inn Road  
London WC1X 8HL  
United Kingdom  

February 7, 2012  

Dear IMPS Editorial Team,  

Thank you for this copy edit. There was only one change that we did not accept in the copy edited manuscript—“We have abbreviated throughout but think using ‘nursing homes’ is more readable so would suggest using in full.”  
  •  We chose to write out this out in full and made the changes throughout.  

We accepted all of the other edits. Further, I identified a paragraph that we had failed to delete previously. I deleted this paragraph and verified the citations to be correct.  

Thank you very much.  

Respectfully,  

Ruth A. Anderson, RN, PhD, FAAN  
Virginia Stone Professor of Nursing  
Research Development Coordinator, Office of Research Affairs  
Senior Fellow, Center for the Study of Aging and Human Development  
Duke University School of Nursing  
Durham, NC 27710  
Office: 919-668-4599  
ruth.anderson@duke.edu